Author’s response to reviews

Title: Identifying High Risk Subgroups of MSM: A Latent Class Analysis using Two Samples

Authors:

M. Kumi Smith (kumi.smith@unc.edu)
Gabriella Stein (ghstein123@gmail.com)
Weibin Cheng (bin_cheng817@163.com)
William Miller (miller.8332@osu.edu)
Joseph Tucker (jdtucker@unc.edu)

Version: 3 Date: 16 Feb 2018

Author’s response to reviews:

Re: INFD-D-17-00815R2

Title: Identifying High Risk Subgroups of MSM: A Latent Class Analysis using Two Samples M. Kumi Smith, MPIA, Ph.D.; Gabriella Stein; Weibin Cheng, M.D., M.S.; William C Miller, M.D., Ph.D., M.P.H.; Joseph D Tucker, M.D., Ph.D.

Journal: BMC Infectious Diseases

February 16, 2018

Dear Dr. Devoto,

We apologize for the second failure of the weblink for our data sources. I have tested the newly inserted link through various means (cutting and pasting, hyperlink, opening on different computers) and am hopeful that this time it should work.

We have uploaded a new version of the manuscript with the corrected link through the system. Please let me know if you encounter any other questions.
Best,

M Kumi Smith
University of North Carolina Chapel Hill